Author's response to reviews

Title: Geographic variation in the choice of adjuvant treatments for women diagnosed with screen-detected breast cancer in Queensland

Authors:

Jeff C Hsieh (jeff.freedom@gmail.com)
Susanna M Cramb (SusannaCramb@cancerqld.org.au)
James M McGree (James.mcgree@qut.edu.au)
Nathan AM Dunn (Nathan.Dunn@health.qld.gov.au)
Peter D Baade (PeterBaade@cancerqld.org.au)
Kerrie L Mengersen (K.Mengersen@qut.edu.au)

Version: 2
Date: 13 March 2015

Author's response to reviews:

1. All authors email address has been display on the Title page as footnote (except the corresponding author).
2. Line numbers have been included in the manuscript using LaTex command supply by the BMC Tex template.
3. On page 3 line 7-9, the sentence "Since only de-identified data was provided by the data custodians and subsequently used in these analyses, no patient consent was obtained" has been added.